Author’s response to reviews

Title: Learning about how Public Involvement strengthens HIV research as a Medical Student

Authors:

Joseph Lewis (jrl221@exeter.ac.uk)

Version: 2 Date: 19 Apr 2020

Author’s response to reviews:

I have added a citation from the newly published PROUD article